Author's response to reviews

Title: An Innovative Model for Video-based on-Ward Supervision - A Descriptive Study

Authors:

Jan B Groener (Jan.Groener@med.uni-heidelberg.de)
Till J Bugaj (TillJohannes.Bugaj@med.uni-heidelberg.de)
Roberta Scarpone (roby1690@outlook.com)
Ansgar Koechel (Ansgar.Koechel@med.uni-heidelberg.de)
Jan Stiepak (Jan.Stiepak@med.uni-heidelberg.de)
Sylvie Branchereau (Sylvie.Branchereau@med.uni-heidelberg.de)
Markus Krautter (Markus.Krautter@med.uni-heidelberg.de)
Wolfgang Herzog (Wolfgang.Herzog@med.uni-heidelberg.de)
Christoph Nikendei (Christoph.Nikendei@med.uni-heidelberg.de)

Version: 2 Date: 31 October 2014

Author's response to reviews: see over
Manuscript ‘An Innovative Model for Video-based on-Ward Supervision – A Descriptive Study’ for publication in BioMed Central.

Dear Mr Ulep,

Thank you very much for your reply to our first version of the manuscript. As requested we included line and page numbers in our revised version. We appreciate you considering our article for publication and look forward to hearing from you soon.

Kind regards

Christoph Nikendei